Author’s response to reviews

Title: Investigation of the anti-cataractogenic mechanisms of curcumin through in vivo and in vitro studies

Authors:
Tao Wang (taowangdu@163.com)
Jing Cao (caojing11984@163.com)
Meng Wang (wangmeng11984@163.com)

Version: 5 Date: 07 Feb 2018

Author’s response to reviews:

Comment1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

Response: Thanks very much for your kind suggestion. We have sent our paper to a commercial language editing company of Editage (https://www.editage.cn/), and the language of the whole manuscript has been edited by a native speaker. We really hope the language is now qualified for publication.